Author’s response to reviews

Title: Work-related musculoskeletal disorders among Nigerian Physiotherapists

Authors:

Adegoke BOA (badegoke@comui.edu.ng)
Akodu AK (akoduashiyatkehinde@yahoo.co.uk)
Oyeyemi AL (alaoyeyemi@yahoo.com)

Version: 8 Date: 12 August 2008

Author’s response to reviews:

The city where the University is located and where the study was conducted is 'Ibadan' and not 'Idban' or 'Ibdan' as written on the title page.

'Aged below 30 years' in the abstract was changed to 'younger than 30 years'.